


TI weekly summary
Week 49



Wednesday 02/12/2020

12/8/2020 Document reference 3

Due to the IT intervention of Monday, impact synchronization were out 

of order Tuesday around 16:50.

Wednesday morning most of people were not allowed to enter in 

machines,

Problem has been fixed during the morning



Thursday 03/12/2020

12/8/2020 Document reference 4

BEQ2 and BEQ3 (SVC for the SPS machine network) switched ON

Friday 04/12/2020

MEQ59 SVC (Meyrin) switched OFF

MEF9 Filter (Meyrin) switched ON

Nb: MEF9 will be used until Christmas break (during MEQ59 repair)



Sunday 06/12/2020 17:58

12/8/2020 Document reference 5

Evacuation alarm in BA2 due to a human error during works for the 

SPS access control system.

Fire Brigade quickly acknowledged the alarm

Monday 07/12/2020 09:00

BEF4 Stable Filter (Prevessin network) switched ON



Intervention forecast

12/8/2020 Document reference 6

Tuesday 08/12 afternoon

- Switch ON of the LHC8 TCR

Wednesday 09/12

- Switch ON of the LHC6 SVC

Thursday 10/12 between 9am and 12am :

- IT would like to deployed a new TN router in the CCR

- A short cut-out of 3 second will append for the SPS tunnel 
connectivity

- to be validated by the TIOC (impact 166117)

This week or next week (will be see during the TIOC of tomorrow)

- Switch OFF of the LHC1 66kV to dig close to the cable (water leak)

- SM18 and LHC1 will be powered from the BE9 (LHC loop)

https://impact.cern.ch/impact/secure/?place=editActivity:166117

